

GOPEN ACCESS

Citation: Kanda E, Epureanu BI, Adachi T, Kashihara N (2023) Machine-learning-based Web system for the prediction of chronic kidney disease progression and mortality. PLOS Digit Health 2(1): e0000188. https://doi.org/10.1371/journal. pdig.0000188

Editor: Syed Sibte Raza Abidi, Dalhousie University, CANADA

Received: July 11, 2022

Accepted: December 20, 2022

Published: January 18, 2023

Copyright: © 2023 Kanda et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data cannot be fully shared publicly. The reasons are as follows: Data contain potentially sensitive information; Patients did not provide informed consent regarding release of personal data; the Ethics Board of Kawasaki Medical School imposed data restriction. The data are owned by Kawasaki Medical School Hospital. Interested readers may request the data at Kawasaki Medical School; URL (Japanese), https://h.kawasaki-m.ac.jp/data/ contact/each. And the following Email address of **RESEARCH ARTICLE**

Machine-learning-based Web system for the prediction of chronic kidney disease progression and mortality

Eiichiro Kanda¹⁰*, Bogdan Iuliu Epureanu²⁰, Taiji Adachi³⁰, Naoki Kashihara⁴⁰

1 Medical Science, Kawasaki Medical School, Kurashikishi, Okayamaken, Japan, 2 College of Engineering, University of Michigan, Ann Arbor, Michigan, United States of America, 3 Department of Biosystems Science, Institute for Life and Medical Sciences, Kyoto University, Kyotoshi, Kyotofu, Japan, 4 Department of Nephrology and Hypertension, Kawasaki Medical School, Kurashikishi, Okayamaken, Japan

So These authors contributed equally to this work.

* kms.cds.kanda@gmail.com

Abstract

Chronic kidney disease (CKD) patients have high risks of end-stage kidney disease (ESKD) and pre-ESKD death. Therefore, accurately predicting these outcomes is useful among CKD patients, especially in those who are at high risk. Thus, we evaluated whether a machine-learning system can predict accurately these risks in CKD patients and attempted its application by developing a Web-based risk-prediction system. We developed 16 riskprediction machine-learning models using Random Forest (RF), Gradient Boosting Decision Tree, and eXtreme Gradient Boosting with 22 variables or selected variables for the prediction of the primary outcome (ESKD or death) on the basis of repeatedly measured data of CKD patients (n = 3,714; repeatedly measured data, n = 66,981) in their electronic-medical records. The performances of the models were evaluated using data from a cohort study of CKD patients carried out over 3 years (n = 26,906). One RF model with 22 variables and another RF model with 8 variables of time-series data showed high accuracies of the prediction of the outcomes and were selected for use in a risk-prediction system. In the validation, the 22- and 8-variable RF models showed high C-statistics for the prediction of the outcomes: 0.932 (95% CI 0.916, 0.948) and 0.93 (0.915, 0.945), respectively. Cox proportional hazards models using splines showed a highly significant relationship between the high probability and high risk of an outcome (p<0.0001). Moreover, the risks of patients with high probabilities were higher than those with low probabilities: 22-variable model, hazard ratio of 104.9 (95% CI 70.81, 155.3); 8-variable model, 90.9 (95% CI 62.29, 132.7). Then, a Webbased risk-prediction system was actually developed for the implementation of the models in clinical practice. This study showed that a machine-learning-based Web system is a useful tool for the risk prediction and treatment of CKD patients.

the hospital may be useful for readers: hsyomu@med.kawasaki-m.ac.jp.

Funding: EK was funded by the Japan Society for the Promotion of Science (KAKENHI Grant Number JP 22K08346). The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Chronic kidney disease (CKD) patients have high risks of end-stage kidney disease (ESKD) and pre-ESKD death. Although the development of a new artificial intelligence (AI) model is expected to be useful for screening CKD patients at high risk, none of the models developed so far have yet been put into practical use. There are many reasons for the difficulties in developing and applying AI models to clinical settings: (1) AI models often have too many variables, some of which are uncommon, to input in busy clinical settings. (2) The accuracy of AI models to predict risks is not always high when evaluating nontarget patients' prognoses. (3) The codes of AI models do not always match electronic medical records of different hospitals. Therefore, in this study, AI models with a small number of commonly used variables were developed. Then, their performances were rigorously evaluated on subgroups of CKD patients classified on the basis of kidney function, age, and diabetes mellitus. Moreover, we developed and opened a Web-based risk-prediction system using the selected AI models that showed the highest accuracies for clinical practice: http://160.16.88.112:8000/. Our study fills the gap between AI development in research and AI in clinical practice.

Introduction

Chronic kidney disease (CKD) patients have high risks of end-stage kidney disease (ESKD), cardiovascular disease (CVD), and death [1–3]. With the progression of CKD, the risks of these outcomes increase. In addition, comorbidities of CKD, such as CKD-mineral and bone disorder (MBD), renal anemia, and hyperpotassemia, also appear with CKD progression [4–9]. Moreover, the medical costs of CKD therapy are high: CKD imposes heavy physical and economic burdens on patients [10,11]. To slow CKD progression, control of the various risk factors, such as diabetes, hypertension, and proteinuria, is needed [12].

Kidney Disease: Improving Global Outcomes (KDIGO) 2012 Clinical Practice Guideline for the Evaluation and Management of CKD proposed a classification system to diagnose CKD, recommended referral to specialists, and provided models of CKD care [4]. Accurate prognoses of a patient's renal function and life expectancy are useful for diagnosing CKD patients with a high risk of ESKD and determining therapeutic strategies for them [4,5]. Thus, an easily accessible risk-prediction system would be useful for these purposes.

The risk prediction of an outcome, such as ESKD, is evaluated using the probabilities of the outcome occurring, as predicted using clinical indices on the basis of statistical models [13–15]. Probabilities are useful as risk indices in clinical settings for risk estimation, diagnosis, clinical therapy decision, and specialist referral, among others [16]. Machine-learning prediction is also based on probability. Various machine-learning models for the evaluation of CKD progression have been developed [17–21]. Although the development of a new risk-prediction system using machine learning is expected to be useful for CKD therapy, none of these models have yet been put into practical use.

There are many reasons for the difficulties in developing and applying machine-learning models to clinical settings. For example, (1) machine-learning models often have too many variables to input and uncommon variables at clinical settings. (2) The accuracy of machine-learning models to predict risks is not always high when evaluating nontarget patients' prognoses. (3) The code developed for machine learning does not always match electronic medical records (EMRs) of different hospitals. Moreover, developing a risk-prediction system requires

the knowledge of not only medicine but also different fields of data-science engineering, such as machine learning, servers, and networks.

Therefore, this study was conducted to clarify whether the probability of the occurrence of either ESKD or pre-ESKD death in CKD patients can be accurately predicted using machine-learning models with a small number of commonly used variables. Then, the applicability of the models to a subclass of patients was evaluated on the basis of estimated glomerular filtration rate (eGFR), age, and diabetes mellitus (DM) using data from a cohort study of CKD patients. Moreover, we developed a Web-based risk-prediction system for the implementation of the models in clinical practice.

Results

Model selection

Comparison of C-statistics among models. In this study, the models were denoted by their algorithm, dataset type (baseline or time-series dataset), and number of variables (all or v +number). For example, a Random Forest (RF) model based on the baseline dataset including x variables was denoted RF_base_vx. Fig 1A shows the C-statistics of the models used for predicting the primary outcome (ESKD or death) over 1 year. The C-statistics of RF_time_all and RF_time_v8 were higher than 0.85 (p<0.05). The C-statistics for the prediction of ESKD over 1 year showed high accuracy for most of the models (Fig 1B). In particular, the C-statistics of 5 models were more than 0.9 for the prediction over 1 year (p<0.05). Moreover, these models exhibited high C-statistics for the prediction of death over 1 year (Fig 1C). Similarly, the models showed high prediction accuracies for the primary outcome over 2 and 3 years (S1 Fig).

Comparison of model performance in subclasses. The C-statistics of the primary outcome were also compared among models for subclasses (S2 Fig). In each subclass, the models showed high C-statistics. S1 Table summarizes the applicability of the models to patients with various conditions. Most of the models with all the variables provided high C-statistics. In particular, RF_base_all, RF_time_all, and RF_time_v8 showed high performances. Considering these results and the calculation speed of the server, RF_time_all and RF_time_v8 were selected as representative models for the development of a Web-based risk-prediction system.

The relationship between the risk of the primary outcome and its predicted probabilities was analyzed (Fig 2). The hazard ratios (HRs) of RF_time_all and RF_time_v8 were associated with high probabilities (p<0.0001). For the models to be informative and show whether a patient was at a high risk, the cutoff levels of the predicted probability were 8.9% for RF_time_all and 8.2% for RF_time_v8.

Validation of the machine-learning models using CKD cohort study data

Baseline characteristics and outcome events of the validation dataset are shown in <u>S2 Table</u> and <u>S3 Table</u>. The mean \pm SD and median (IQR) of the probability predicted by the models were 2.5 \pm 5.6, 1.0 (0.6, 1.9)% for RF_time_all and 2.2 \pm 6.0, 0.6 (0.3, 1.3)% for RF_time_v8.

S4 Table summarizes the applicability of the models to patients with various conditions. RF_time_all and RF_time_v8 showed high C-statistics in the cohort study data. Their high C-statistics for the prediction of the primary outcome over 3 years were 0.932 95% confidence intervals (CIs) (0.916, 0.948) and 0.93 (0.915, 0.945). These values were statistically significantly higher than that using obtained the kidney-failure-risk equation (KFRE) 0.905 (0.881, 0.929; p = 0.035, 0.033, respectively).

For the prediction of ESKD over 3 years, the models showed high C-statistics of more than 0.95: 0.966 (95% CI 0.956, 0.976), 0.966 (95% CI 0.957, 0.975). For the prediction of death over



Fig 1. Model performances for prediction in outcomes over 1 year at model selection stage. Figures show the C-statistics for the prediction of outcomes with 95% CIs of the models. A: Primary outcome. B: ESKD. C: Death. Abbreviations: ESKD, end-stage kidney disease; eGFR, estimated glomerular filtration rate; UPCR, urinary protein-to-creatinine ratio; KFRE, kidney failure risk equation; RF, Random Forest; GB, Gradient Boosting Decision Tree; XG, eXtreme Gradient Boosting; LRM, logistic regression model; CI, confidence interval.

https://doi.org/10.1371/journal.pdig.0000188.g001

3 years, the C-statistics of the models were higher than 0.8: 0.859 (95% CI 0.821, 0.898), 0.854 (95% CI 0.817, 0.89).

Cox proportional hazards models showed the relationships between high probabilities predicted using the models and high risks of the primary outcome on the basis of the follow-up period (Fig 3).

Risk of outcome events and high-risk patients. Among the patients, 5.5% were classified into the high-risk groups on the basis of the cutoff levels of RF_time_all and RF_time_v8, and they showed different characteristics from the low-risk group (Table 1).

Survival probabilities of the high-risk groups classified on the basis of RF_time_all and RF_time_v8 were lower than those of the low-risk groups (Table 2). The HRs of the primary outcome of the high-risk groups classified on the basis of RF_time_all and RF_time_v8 were 104.9 (95% CI 70.81, 155.3) and 90.9 (95% CI 62.29, 132.7), respectively. Likewise, the HRs of ESKD were 103.3 (95% CI 70.0, 152.4) and 89.6 (95% CI 61.6, 130.3), and the HRs of death were 14.85 (95% CI 9.75, 22.62) and 14.36 (95% CI 9.43, 21.86) for RF_time_all and RF_time_v8, respectively.



Fig 2. Relationships between risk of primary outcome and its predicted probabilities at model selection stage. The figure shows the relationships between the risk of the primary outcome and its predicted probabilities. Histograms of the probabilities and the log hazards of the primary outcome are shown in the upper and lower panels, respectively. A: RF_time_all. B: RF_time_v8. Abbreviation: RF, Random Forest.

https://doi.org/10.1371/journal.pdig.0000188.g002

Implementation of the models in clinical settings

We developed a Web-based risk-prediction system using RF_time_all and RF_time_v8. Then, we uploaded the system on the Web site for users to access and easily estimate CKD patients' risk (http://160.16.88.112:8000/) (Fig 4).

Discussion

In this study, we developed a Web-based risk-prediction system consisting of machine-learning models for the prediction of CKD patients' renal and life prognoses. The machine-learning-based probability could accurately determine CKD patients' prognoses, and the system was applicable with high accuracy to CKD patients with various conditions and comorbidities. Our findings indicated that the machine-learning-based probability can be a new risk index for CKD patients. Moreover, we developed a Web-based risk-prediction system using the two selected RF models to apply the predicted probability as an index in clinical practice. This system enabled easy and quick estimation of the prognoses of CKD patients, and it provided a recommendation that patients be referred to a specialist. To the best of our knowledge, this study is the first on the risk prediction of CKD patients using artificial intelligence (AI), whose performances were thoroughly tested and implemented. This would be a supportive tool for providing medical treatment to patients at high risks of outcome events.

Probabilities of the patients' prognoses are useful as a medical index in clinical practice. Various risk-prediction models have been developed using statistical models to predict CKD progression [13–15]. KFRE was developed for CKD patients using Cox proportional hazards models, and the survival index was developed using logistic regression (LR) models by Dialysis Outcomes and



Fig 3. Relationships between risk of primary outcome and its predicted probability. Histograms of the probabilities and the hazard ratios of the primary outcome are shown in the upper and lower panels, respectively. A, RF_time_all, p<0.0001; B, RF_time_v8, p<0.0001. Abbreviations: HR, hazard ratio; RF, Random Forest.

https://doi.org/10.1371/journal.pdig.0000188.g003

Practice Patterns Study (DOPPS) for hemodialysis patients [16,22]. These probabilities can extract patients' features as a single variable, such as a clinical propensity score, and make it easy to estimate the condition of patients by simply substituting values in the models.

Treatment plans are often selected while considering numerous trade-offs. For example, should a low-protein diet be prescribed to prevent hyperphosphatemia and CKD-MBD, or should it not be prescribed to prevent malnutrition/sarcopenia/frailty? In this situation, probabilities of patients' prognoses will give us the candidate strategies and assist in the selection of suitable therapies [16]. Probabilities estimated using machine-learning models, which allow a much larger number of variables than statistical models, can be applied to a highly complicated situation involving follow-up conditions, drug selection, the determination of blood access for dialysis, and the initiation of dialysis and kidney transplantation. Probability-based machine-learning models have a wide range of applications in medicine.

It is often very difficult for medical staff to calculate statistical formulae including log or exponential functions with many indices during busy medical examinations. To enable busy medical staff and patients to access the system quickly and easily, our risk-prediction system was uploaded on a Web site. The system developed in this study was based on RF models in a server, which analyzed patient data at a high speed and could be easily accessed through the internet using PCs and smartphones. To decrease the burden of measurement and inputting patients' data, the variables used in the models were universally assessed in clinical settings, and the number of variables in the models of our system was limited. AI and communication technologies were used to realize this system.

Our models are applicable to clinical settings within their limitations. First, the data were obtained in Japan. The accuracies of the models for Asian patients will be high, but the results

Table 1.	Baseline characteristics of 1	isk groups classifie	d on the basis of the resul	ts of machine-learning models.
		0 1		

Model names		RF_time_all		RF_time_v8		
Risk groups	Low-risk group	High-risk group	<i>p</i> value	Low-risk group	High-risk group	p value
N (%)	25,422 (94.5)	1,484 (5.5)		25,428 (94.5)	1,478 (5.5)	
Demographic characteristics						
Age (years)	61.2±16.3	62.0±17.9	0.056	61.2±16.3	62.2±18.1	0.022
Male (%)	13,000 (51.1)	779 (52.5)	0.31	13,020 (51.2)	759 (51.4)	0.92
Comorbidities						
DM (%)	5,305 (20.9)	332 (22.4)	0.17	5,293 (20.8)	344 (23.3)	0.025
Hypertension (%)	5,234 (20.6)	336 (22.6)	0.060	5,230 (20.6)	340 (23.0)	0.027
CVD (%)	159 (0.6)	30 (2.0)	< 0.0001	162 (0.6)	27 (1.8)	< 0.0001
Laboratory data						
eGFR (mL/min/1.73m ²)	73.3±31.0	68.7±26.4	< 0.0001	73.3±31.0	69.2±29.7	< 0.0001
Albumin (g/dL)	4.2±0.5	4.0±0.63	< 0.0001	4.2±0.5	4.0±0.6	< 0.0001
Sodium (mmol/L)	140.3±2.6	139.9±3.0	< 0.0001	140.3±2.6	139.9±300	< 0.0001
Potassium (mmol/L)	4.2±0.4	4.2±0.5	0.74	4.2±0.4	4.2±0.5	0.16
Calcium (mg/dL)	9.2±0.5	9.0±0.7	< 0.0001	9.2±0.5	9.1±0.7	< 0.0001
Phosphorus (mg/dL)	3.4±0.7	3.5±1.0	0.028	3.4±0.7	3.5±0.9	0.008
LDL (mg/dL)	111.0±32.1	110.6±36.5	0.71	111.0±32.1	110.0±36.7	0.32
Uric acid (mg/dL)	5.2±1.5	5.3±1.6	0.54	5.2±1.5	5.3±1.6	0.40
WBC (10 ³ /µL)	6.3±3.0	6.6±4.1	< 0.0001	6.3±3.0	6.6±4.1	< 0.0001
Hemoglobin (g/dL)	13.6±1.8	13.1±2.3	< 0.0001	13.6±1.8	13.1±2.3	< 0.0001
UPCR (g/gCre)	0.27 [0.11, 1.00]	0.43 [0.14, 2.47]	<0.0001	0.27 [0.11, 1.00]	0.42 [0.14, 2.42]	<0.0001
Medications						
RAASI (%)	4,407 (17.3)	268 (18.1)	0.48	4,404 (17.3)	271 (18.3)	0.32
Phosphorus absorbent (%)	209 (0.8)	52 (3.5)	< 0.0001	214 (0.8)	47 (3.2)	< 0.0001
Vitamin D (%)	653 (2.6)	63 (4.2)	< 0.0001	653 (2.6)	63 (4.3)	< 0.0001
Statin (%)	4,162 (16.4)	180 (12.1)	< 0.0001	4,145 (16.3)	197 (13.3)	0.002
Uric-acid-lowering medicines (%)	2,272 (8.9)	174 (11.7)	< 0.0001	2,286 (9.0)	160 (10.8)	0.020
ESA (%)	665 (2.6)	82 (5.5)	<0.0001	675 (2.7)	72 (4.9)	< 0.0001

Continuous variables are shown as mean±SD or median (interquartile range). Categorical variables are shown as n (%).Abbreviations: DM, diabetes mellitus; CVD, cardiovascular disease; eGFR, estimated glomerular filtration rate; LDL, low-density lipoprotein; WBC, white blood cells; UPCR, urinary protein-to-creatinine ratio; RAASI, renin-angiotensin-aldosterone system inhibitor; ESA, erythropoietin-stimulating agent.

https://doi.org/10.1371/journal.pdig.0000188.t001

Table 2. C	Outcome events of ris	k groups	classified	on the basis (of machine-	learning models.
------------	-----------------------	----------	------------	----------------	-------------	------------------

Model names		RF_time_all		RF_time_v8			
Risk groups	Low-risk group	High-risk group	<i>p</i> value	Low-risk group	High-risk group	<i>p</i> value	
Primary outcome (%)	235 (0.9)	43 (2.9)	< 0.0001	238 (0.9)	40 (2.7)	< 0.0001	
ESKD (%)	153 (0.6)	34 (2.3)	< 0.0001	156 (0.6)	31 (2.1)	< 0.0001	
Death (%)	82 (0.3)	9 (0.6)	0.099	82 (0.3)	9 (0.6)	0.099	
Follow-up period (days)	901 [527, <u>1</u> ,052]	710 [407, 1,038]	<0.0001	898 [526, 1052]	738.5 [428, 1043]	<0.0001	

Continuous variables are shown as median (interquartile range). Categorical variables are shown as n (%). Abbreviations: ESKD, end-stage kidney disease.

https://doi.org/10.1371/journal.pdig.0000188.t002



D

Fig 4. Images of Web pages of risk-prediction system. A. Conceptual design of the system: Patients' data are input to the machine-learning models. The results are shown on the browser. B. Start page: Go to the Web site, <u>http://160.16.88.112:8000/</u>. C. Input page: Input data of a patient. D. Result page: The expected probability is shown with messages to focus the user's attention on CKD. Abbreviation: CKD, chronic kidney disease.

https://doi.org/10.1371/journal.pdig.0000188.g004

obtained in non-Asian countries may be biased by the patient population. Transfer learning will improve the accuracy of this system for global use as described above. Second, the data did not sufficiently include those for assessing variables that were not included in EMRs, such as lifestyle, food intake, family history, and social determinants of health. Third, new drugs that can effectively prevent CKD progression, such as sodium-glucose cotransporter-2 inhibitors,

have been developed [23,24]. In this study, we did not use data on those drugs, because it was only a year since it was approved for use in CKD therapy in Japan. It is necessary to wait until sufficient data on those drugs will be accumulated. Fourth, RF-related models were developed in this study. A deep learning model was not used because the number of variables was small and because deep learning models tend to overfit. Thus, RF-related models were suitable for this study. Fifth, EMR data were used in this study. Although such data reflect real-world settings, all of the clinical variables are not always measured. Variables rarely measured have many missing values and are difficult to use in machine-learning models. Moreover, there is a problem of when to measure variables with day-to-day variations, such as blood glucose level. Therefore, commonly used and stable variables which were used in this study. It is necessary to examine whether variables are appropriate for model development when real-world data are used. Sixth, the WEB system was developed in this observational study. An interventional study should be conducted to show the effects of this system on CKD therapy and improvement of CKD progression.

In conclusion, this study showed that the machine-learning-based Web system accurately predicts a CKD patient's increased risk of ESKD or death and is useful as a new tool for clinical practice.

Materials and methods

This study consisted of 4 steps; (1) model development, (2) model selection, (3) model validation, and (4) development of the WEB-based risk-prediction system. Steps (1) and (2) were conducted using data from EMRs of CKD patients. Step (3) was conducted using data from a cohort study of CKD patients.

Study design and ethics

In this study, the machine-learning models were developed using data from EMRs of Kawasaki Medical School Hospital, Okayama, Japan, from January 1st, 2014 to December 31st, 2017 and validated using data from a cohort study of CKD patients who visited Kawasaki Medical School Hospital from January 1st, 2018 to December 31st, 2020 (CKD cohort study). This study was approved by the ethics committee of Kawasaki Medical School. Owing to the nature of this study, it was exempt from the need to obtain informed consent from participants (Nos. 3444, 3445, and 5306). The study was performed in accordance with the relevant guidelines and the Declaration of Helsinki of 1975 as revised in 1983.

Step (1) Development of machine-learning models

Study population. The EMR data as a dataset were extracted from the server in Kawasaki Medical School Hospital. The dataset was preprocessed for analysis because it was composed of all data of all patients in a column. The dataset was divided by variables, and the separated data were merged for each patient and medical examination day in a table for analysis. The dataset contained time-series data of patients (time-series dataset n = 83,703) (S3 Fig).

The exclusion criteria were as follows: patients younger than 20 years, patients on any type of dialysis, and patients who had malignancies. The patients with malignancies often show a shorter survival than others and tend to die before they reach ESKD. Their disease courses are different from the typical course of CKD. If their data, which are not related to CKD, had been included among the data for machine learning, they may affect the performance of machine-learning models and variable selection. Therefore, patients with malignancies were excluded in this study. Moreover, patients who had missing baseline data were excluded from the data-set. Thus, the time-series data obtained on every medical examination day from 3,714 patients

were used in the development of machine-learning models (time-series dataset n = 66,981). The median (IQR) number of repeatedly measured times per patient was 16 (7, 29). The first-visit data of a patient were treated as baseline data (baseline dataset n = 3,714).

In this study, two types of machine-learning model were developed using baseline data or time-series data. To develop and select the models, the data were further assigned into the development dataset (including 80% of the patients, baseline dataset n = 2,967, time-series dataset n = 31,763) and a selection dataset (baseline dataset n = 747) on the basis of patient ID number.

The development dataset contained baseline data and time-series data, each of which was further divided into two datasets on the basis of patient ID number. About 70% of the development dataset was used to train the models (baseline dataset n = 2,076, time-series dataset n = 22,234), and the rest was used to test the models (baseline dataset n = 891, time-series dataset n = 9,529). As a result, four datasets were prepared: the baseline dataset to train the models (baseline dataset n = 22,234), the baseline dataset to test the models (baseline dataset to train the models (baseline dataset n = 22,234), the baseline dataset to test the models (baseline dataset n = 22,234), the baseline dataset to test the models (baseline dataset n = 891), and the time-series dataset to test the models (baseline dataset n = 891), and the time-series dataset to test the models (time-series dataset n = 9,529). The two training datasets included the same patients as well as the two testing datasets.

On the other hand, the selection dataset contained only baseline data (baseline dataset n = 747). Baseline characteristics and outcome events are shown in <u>S5 Table</u> and <u>S6 Table</u>, respectively.

Variables. The variables were CKD-related factors and medications in the EMR data [4,25–28]. They were selected for global use on the basis of their being common measurements used at outpatient clinics and ease of data input by users at implementation. The variables with more than 20% of their time-series data missing were not included in the analysis, because these variables were considered not to be usually measured at clinical settings.

The following variables were used in the analysis as follows (S7 Table): age; gender; DM; hypertension; history of CVD; eGFR; serum albumin, sodium, potassium, calcium, phosphorus, low-density lipoprotein, and uric acid levels; white blood cell count; hemoglobin level; urinary protein-to-creatinine ratio (UPCR); and use of renin-angiotensin-aldosterone system inhibitors, phosphorus absorbents, vitamin D, statins, uric-acid-lowering medicines, and erythropoietin-stimulating agents. The primary outcome was ESKD or death. If no outcomes were observed within the follow-up period, the observation data were treated as censored data. The onset of ESKD was defined as the initiation of renal replacement therapy. None of the patients in this study had received a kidney transplantation.

eGFR was calculated using the equation for the Japanese population on the basis of the serum creatinine level measured using the enzymatic method [29]. The baseline data were defined as the first measurements within 30 days of the initial eGFR. KFRE was calculated using eGFR, age, gender, and natural logarithm (urine albumin-to-creatinine ratio, UACR) [14]. Negative values of KFRE were treated as zero. Because the Japanese national health insurance covers UACR measurement only for DM patients, UACR was estimated from the UPCR using Weaver *et al.*'s estimation formulae [30]. The last observation data carried forward were input in place of all missing time-series data.

Machine-learning models. The *scikit-learn* and *numpy* packages of Python were mainly used in model development. RF, Gradient Boosting Decision Tree (GB), and eXtreme Gradient Boosting (XG) models were developed using *RandomForestClassifier*, *GradientBoostingClassifier*, and *XGBClassifier* in *scikit-learn*, respectively. Their parameters were determined using a grid search algorithm, *GridSearchCV*, to search for the hyperparameters for the models, which were the number of estimators (trees), the maximum percentage of features used in each tree, maximum depth, and class weight. The parameters of *GridSearchCV* were the maximum number of trees of 1000, depth of 10, leaf of 10, and sample split of 10. The parameters of the selected models

were as follows. RF_time_all: number of trees = 1000, maximum depth = 10, minimum samples leaf = 5, minimum samples split = 5; RF_time_v8: number of trees = 1000, max depth = 10, minimum samples leaf = 5, minimum samples split = 5. In the development of the models with selected variables, the variables were ranked by their permutation importance in each model using *numpy* and then selected. *pickle* was used to save and load these trained models.

Model development. We developed risk-prediction models to classify the primary outcome over 1 year using the RF, GB, and XG models, which were trained to predict the primary outcome. Each type of machine-learning model included all 22 variables or selected variables. Variables were selected by their permutation importance in each model (S4 Fig). The numbers of variables selected were 7 for RF_base, 8 for RF_time, 13 and 6 for GB_base, 7 for GB_time, 17 and 7 for XG_base, and 10 and 6 for XG_time.

A multivariate LR model was developed using the hierarchical backward elimination procedure. The LR model with 11 selected variables was developed (<u>S8 Table</u>).

Statistical analyses. Baseline characteristics and outcomes were compared between the development and selection datasets using the chi-square test, t-test, and Mann-Whitney U test, as appropriate. After the proportional hazards assumption was confirmed using a double logarithmic plot, multivariate Cox proportional hazards models were used to compare the risks of outcomes between groups. The results are presented here as HRs with 95% CIs. In the analysis of the competing risks of ESKD and death, we used Fine and Gray competing risk regression models. Statistical significance was defined as a two-sided p<0.05. All statistical analyses were carried out using SAS version 9.4 (SAS Institute, Cary, NC, USA), Python 3.8.2 (Python Software Foundation, DE, USA), and R version 3.6.1 (R Foundation for Statistical Computing, Vienna, Austria).

Step (2) Model selection

Dataset and accuracies of the risk prediction

After the models were developed, they were applied to the model selection dataset including the baseline data of 747 patients. To evaluate models, a dataset composed of 1,000 sampled datasets was generated from the model selection dataset by bootstrapping 1,000 times so that the C-statistics with 95% CIs for the prediction of the outcomes were evaluated.

Moreover, subgroup analyses were conducted. The C-statistics were assessed in the following relevant subgroups: high (60 mL/min/1.73m² or higher) and low (less than 60 mL/min/ 1.73m²) eGFRs, DM status, and young (less than 65 years) and old (65 years or older) ages.

Risks of outcomes. A well-developed model is one for which the predicted probabilities strongly correlate with the risks of the outcomes. To evaluate the relationship between the probability predicted using the models and the risk of the primary outcome, univariate Cox proportional hazards models including the spline terms of the predicted probability were developed using the model selection dataset.

Moreover, for users to easily understand the severity of risks in a patient, cutoff levels of the predicted probabilities were determined for the categorization of patients. A cutoff level of each model was determined on the basis of the Youden index on the receiver operating characteristic curve of the prediction of the primary outcome over 1 year using the development dataset [31]. Then, the patients were categorized into high- and low-risk groups.

Step (3) Validation of the machine-learning models using data from a cohort study

Study population. The machine-learning models were validated using a dataset of a cohort study different from the development dataset. The CKD cohort study was conducted

for three years. CKD patients who regularly visited the outpatient clinic of Kawasaki Medical School were registered (n = 67,957) (S5 Fig). The exclusion criteria were the same as those for the dataset used for the model development. Moreover, the patients who had no eGFR data were also excluded. A total of 26,906 patients were included in the analysis and followed up during the study period. The data were extracted from the cohort dataset derived from an EMR server in Kawasaki Medical School Hospital. The data at the first visit of patients during the study period were treated as baseline data. The dataset contained baseline data only.

Variables. The names and definitions of the variables were the same as those for the dataset used for the model development (S7 Table). Multiple imputation was conducted to account for missing data in analyses. The primary outcome was defined as ESKD or death for the follow-up period. The data of patients who did not have either of these outcomes were treated as censored data. The probability of the primary outcome was estimated using the machinelearning models and used as a risk index.

Statistical analysis. The C-statistics and 95% CIs for the accuracies of the prediction of the outcomes, such as the primary outcome, ESKD, or death, of the estimated probability were evaluated using the 1000-times bootstrap method. The C-statistics of the models for the outcomes over 1 year were statistically compared with that of KFRE using the bootstrap method.

For the probabilities to be used in clinical settings, not only incident risk but also survival analysis considering a long follow-up period was required. Thus, univariate Cox proportional hazards models including the spline terms of the probability were used to evaluate the relationship between the probability and risk of the primary outcome.

Then, the patients were divided into two groups on the basis of the cutoff levels determined at Step (2). The hazard of the primary outcome in the high-risk group was compared with that in the low-risk group. In the analysis of the competing risks of ESKD and death, Fine and Gray competing risk regression models were examined.

Step (4) Development of a Web-based risk-prediction system and its specifications

To deploy the machine-learning models on the Web for users to try, we developed a Webbased risk-prediction system. We set up a Web server and developed a Web system in which the models were applied at the backend of the system. The system was used as follows. (1) A user accesses the system through a Web browser on a PC or smartphone. (2) Entered user data are applied to machine-learning models through the server, which estimates the probability of the primary outcome. (3) The results and messages to the user are shown on the Web browser.

In this system, there are several formulae with which users can easily calculate the probability of the primary outcome. eGFR was calculated using the Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI) equation for white and black people, and using the equation for the Japanese population for Asian people on the basis of the serum creatinine level measured using the enzymatic method [29,32]. This system minimizes prediction errors due to differences in the methods of measuring creatinine levels, because the risk probability was estimated on the basis of eGFR. Moreover, units can be selected for patient data, because the units of variables used in different countries vary.

Supporting information

S1 Fig. C-statistics of models for prediction of outcomes over 2 and 3 years at model selection stage.

(PDF)

S2 Fig. C-statistics of models for prediction of primary outcome over 1 year in subclasses at model selection stage. (PDF) S3 Fig. Study population for model development and selection. (PDF) S4 Fig. Ranks of variables in models. (PDF) S5 Fig. Study population for model validation. (PDF) S1 Table. Summary of C-statistics of models at model selection stage. (PDF) S2 Table. Baseline characteristics of patients used as dataset for model validation. (PDF) S3 Table. Outcome events of model validation dataset. (PDF) S4 Table. Summary of model performances at model validation stage. (PDF) S5 Table. Baseline characteristics of model development and selection datasets. (PDF) S6 Table. Outcome events of model development and selection datasets. (PDF) **S7** Table. Definition of variables. (PDF) S8 Table. Selected variables in logistic regression model and effect of each variable on primary outcome. (PDF)

Author Contributions

Conceptualization: Eiichiro Kanda, Bogdan Iuliu Epureanu. Data curation: Eiichiro Kanda, Taiji Adachi. Formal analysis: Eiichiro Kanda, Bogdan Iuliu Epureanu. Funding acquisition: Eiichiro Kanda. Investigation: Eiichiro Kanda, Bogdan Iuliu Epureanu, Taiji Adachi. Methodology: Eiichiro Kanda, Bogdan Iuliu Epureanu, Taiji Adachi. Project administration: Eiichiro Kanda, Naoki Kashihara. Resources: Eiichiro Kanda, Bogdan Iuliu Epureanu, Naoki Kashihara. Software: Eiichiro Kanda, Bogdan Iuliu Epureanu. Supervision: Eiichiro Kanda, Naoki Kashihara. Visualization: Eiichiro Kanda, Taiji Adachi, Naoki Kashihara.

Writing – original draft: Eiichiro Kanda, Bogdan Iuliu Epureanu.

Writing – review & editing: Eiichiro Kanda, Bogdan Iuliu Epureanu, Taiji Adachi, Naoki Kashihara.

References

- Gansevoort RT, Correa-Rotter R, Hemmelgarn BR, Jafar TH, Heerspink HJ, Mann JF, et al. Chronic kidney disease and cardiovascular risk: epidemiology, mechanisms, and prevention. Lancet. 2013; 382: 339–352. https://doi.org/10.1016/S0140-6736(13)60595-4 PMID: 23727170
- Matsushita K, van der Velde M, Astor BC, Woodward M, Levey AS, de Jong PE, et al. Association of estimated glomerular filtration rate and albuminuria with all-cause and cardiovascular mortality in general population cohorts: a collaborative meta-analysis. Lancet. 2010; 375: 2073–2081. <u>https://doi.org/ 10.1016/S0140-6736(10)60674-5 PMID: 20483451</u>
- Astor BC, Matsushita K, Gansevoort RT, van der Velde M, Woodward M, Levey AS, et al. Lower estimated glomerular filtration rate and higher albuminuria are associated with mortality and end-stage renal disease. A collaborative meta-analysis of kidney disease population cohorts. Kidney Int. 2011; 79: 1331–1340. https://doi.org/10.1038/ki.2010.550 PMID: 21289598
- KDIGO. Clinical Practice Guideline for the Evaluation and Management of Chronic Kidney Disease. Kidney int. 2013; 3: 1–150.
- Japanese Society of Nephrology. Essential points from Evidence-based Clinical Practice Guidelines for Chronic Kidney Disease 2018. Clin Exp Nephrol. 2019; 23: 1–15. <u>https://doi.org/10.1007/s10157-018-1648-1</u> PMID: 30506489
- Matsushita K, Ballew SH, Wang AY, Kalyesubula R, Schaeffner E, Agarwal R. Epidemiology and risk of cardiovascular disease in populations with chronic kidney disease. Nat Rev Nephrol. 2022; 18: 696– 707. https://doi.org/10.1038/s41581-022-00616-6 PMID: 36104509
- Nakagawa N, Sofue T, Kanda E, Nagasu H, Matsushita K, Nangaku M, et al. J-CKD-DB: a nationwide multicentre electronic health record-based chronic kidney disease database in Japan. Sci Rep. 2020; 10: 7351. https://doi.org/10.1038/s41598-020-64123-z PMID: 32355258
- Sofue T, Nakagawa N, Kanda E, Nagasu H, Matsushita K, Nangaku M, et al. Prevalence of anemia in patients with chronic kidney disease in Japan: A nationwide, cross-sectional cohort study using data from the Japan Chronic Kidney Disease Database (J-CKD-DB). PLoS One. 2020; 15: e0236132. https://doi.org/10.1371/journal.pone.0236132 PMID: 32687544
- Kashihara N, Kohsaka S, Kanda E, Okami S, Yajima T. Hyperkalemia in Real-World Patients Under Continuous Medical Care in Japan. Kidney Int Rep. 2019; 4: 1248–1260. <u>https://doi.org/10.1016/j.ekir.</u> 2019.05.018 PMID: 31517144
- GBD Chronic Kidney Disease Collaboration. Global, regional, and national burden of chronic kidney disease, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. Lancet. 2020; 395: 709–733. https://doi.org/10.1016/S0140-6736(20)30045-3 PMID: 32061315
- Olufade T, Lamerato L, Sánchez JJG, Jiang L, Huang J, Nolan S, et al. Clinical Outcomes and Healthcare Resource Utilization in a Real-World Population Reflecting the DAPA-CKD Trial Participants. Adv Ther. 2021; 38: 1352–1363. https://doi.org/10.1007/s12325-020-01609-2 PMID: 33474707
- 12. Inaguma D, Imai E, Takeuchi A, Ohashi Y, Watanabe T, Nitta K, et al. Risk factors for CKD progression in Japanese patients: findings from the Chronic Kidney Disease Japan Cohort (CKD-JAC) study. Clin Exp Nephrol. 2017; 21: 446–456. https://doi.org/10.1007/s10157-016-1309-1 PMID: 27412450
- 13. Echouffo-Tcheugui JB, Kengne AP. Risk models to predict chronic kidney disease and its progression: a systematic review. PLoS Med. 2012; 9: e1001344. https://doi.org/10.1371/journal.pmed.1001344 PMID: 23185136
- Tangri N, Stevens LA, Griffith J, Tighiouart H, Djurdjev O, Naimark D, et al. A predictive model for progression of chronic kidney disease to kidney failure. JAMA. 2011; 305: 1553–1559. <u>https://doi.org/10. 1001/jama.2011.451</u> PMID: 21482743
- Tangri N, Kitsios GD, Inker LA, Griffith J, Naimark DM, Walker S, et al. Risk prediction models for patients with chronic kidney disease: a systematic review. Ann Intern Med. 2013; 158: 596–603. https:// doi.org/10.7326/0003-4819-158-8-201304160-00004 PMID: 23588748

- Tangri N, Ferguson T, Komenda P. Pro: Risk scores for chronic kidney disease progression are robust, powerful and ready for implementation. Nephrol Dial Transplant. 2017; 32: 748–751. https://doi.org/10. 1093/ndt/gfx067 PMID: 28499025
- Xiao J, Ding R, Xu X, Guan H, Feng X, Sun T, et al. Comparison and development of machine learning tools in the prediction of chronic kidney disease progression. J Transl Med. 2019; 17: 119. <u>https://doi.org/10.1186/s12967-019-1860-0 PMID: 30971285</u>
- Liu Y, Zhang Y, Liu D, Tan X, Tang X, Zhang F, et al. Prediction of ESRD in IgA Nephropathy Patients from an Asian Cohort: A Random Forest Model. Kidney Blood Press Res. 2018; 43: 1852–1864. <u>https:// doi.org/10.1159/000495818</u> PMID: 30537719
- Zhao J, Gu S, McDermaid A. Predicting outcomes of chronic kidney disease from EMR data based on Random Forest Regression. Math Biosci. 2019; 310: 24–30. https://doi.org/10.1016/j.mbs.2019.02.001 PMID: 30768948
- Ravizza S, Huschto T, Adamov A, Böhm L, Büsser A, Flöther FF, et al. Predicting the early risk of chronic kidney disease in patients with diabetes using real-world data. Nat Med. 2019; 25: 57–59. https://doi.org/10.1038/s41591-018-0239-8 PMID: 30617317
- Kanda E, Kanno Y, Katsukawa F. Identifying progressive CKD from healthy population using Bayesian network and artificial intelligence: A worksite-based cohort study. Sci Rep. 2019; 9: 5082. https://doi. org/10.1038/s41598-019-41663-7 PMID: 30911092
- Kanda E, Bieber BA, Pisoni RL, Robinson BM, Fuller DS. Importance of simultaneous evaluation of multiple risk factors for hemodialysis patients' mortality and development of a novel index: dialysis outcomes and practice patterns study. PLoS One. 2015; 10: e0128652. https://doi.org/10.1371/journal. pone.0128652 PMID: 26030526
- Heerspink HJL, Stefánsson BV, Correa-Rotter R, Chertow GM, Greene T, Hou FF, et al. Dapagliflozin in Patients with Chronic Kidney Disease. N Engl J Med. 2020; 383: 1436–1446. <u>https://doi.org/10.1056/ NEJMoa2024816 PMID: 32970396</u>
- 24. Kanda E, Nangaku M. Are SGLT2 inhibitors a targeted treatment for diabetic kidney disease? Kidney Int. 2019; 96: 8–10. https://doi.org/10.1016/j.kint.2019.04.005 PMID: 31229049
- KDIGO. KDIGO Clinical Practice Guideline for Anemia in Chronic Kidney Disease. Kidney Int. 2012; Supple 2: 279–335.
- KDIGO. KDIGO 2021 Clinical Practice Guideline for the Management of Blood Pressure in Chronic Kidney Disease. Kidney Int. 2021; 99: S1–S87.
- 27. KDIGO. KDIGO 2020 Clinical Practice Guideline for Diabetes Management in Chronic Kidney Disease. Kidney Int. 2020; 98: S1–S115. https://doi.org/10.1016/j.kint.2020.06.019 PMID: 32998798
- KDIGO. KDIGO 2017 Clinical Practice Guideline Update for the Diagnosis, Evaluation, Prevention, and Treatment of Chronic Kidney Disease-Mineral and Bone Disorder (CKD-MBD). Kidney Int Suppl. 2017; 7: 1–59.
- Matsuo S, Imai E, Horio M, Yasuda Y, Tomita K, Nitta K, et al. Revised equations for estimated GFR from serum creatinine in Japan. Am J Kidney Dis. 2009; 53: 982–92. https://doi.org/10.1053/j.ajkd. 2008.12.034 PMID: 19339088
- Weaver RG, James MT, Ravani P, Weaver CGW, Lamb EJ, Tonelli M, et al. Estimating Urine Albuminto-Creatinine Ratio from Protein-to-Creatinine Ratio: Development of Equations using Same-Day Measurements. J Am Soc Nephrol. 2020; 31: 591–601. <u>https://doi.org/10.1681/ASN.2019060605</u> PMID: 32024663
- Akobeng AK. Understanding diagnostic tests 3: Receiver operating characteristic curves. Acta Paediatr. 2007; 96: 644–647. https://doi.org/10.1111/j.1651-2227.2006.00178.x PMID: 17376185
- Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF, Feldman HI, et al. A new equation to estimate glomerular filtration rate. Ann Intern Med. 2009; 150: 604–612. <u>https://doi.org/10.7326/0003-4819-150-9-200905050-00006</u> PMID: 19414839